Author’s response to reviews

Title: Spinocerebellar ataxia type 31 associated with REM sleep behavior disorder: a case report

Authors:

Kazumasa Shindo (kshindo@yamanashi.ac.jp)
Tohko Sato (tohko@yamanashi.ac.jp)
Hiroaki Murata (murata@yamanashi.ac.jp)
Yuta Ichinose (yuta@yamanashi.ac.jp)
Takanori Hata (hata@yamanashi.ac.jp)
Yoshihisa Takiyama (ytakiyama@yamanashi.ac.jp)

Version: 1 Date: 20 Dec 2018

Author’s response to reviews:

Dear Editors of BMC Neurology

The comments of the reviewer have been again helpful in allowing us to revised manuscript (MS No. NURL-D-18-00616). We have attempted to address the questions raised by the referees according to the following:

Editor comments:

The result of dopamine transporter scan was already described in case presentation of the first version of our manuscript. If possible, we would like to add this result as Figure 2. Figure 2 in the first version of our manuscript was changed to Figure 3 and Figure legends were also changed.

Reviewer #1:

1. We rewrote and added the description of “There is the limitation in this study due to a single case report and since the possibility of ∙ ∙ ∙ ∙ ∙ ∙” from page 5, line 23 to page 6, line 1 according to comments from reviewers 1 and 2.
Reviewer #2:

1. We rewrote and added the description of “She was taking two medicines • • •” in page 3, lines 21-23.

2. We rewrote the description of “Electroencephalography revealed no epileptic discharges. PSG study demonstrated that • • •” in page 4, lines 7-12.

3. We rewrote and added the description of “There is the limitation in this study due to a single case report and since the possibility of • • •” from page 5, line 23 to page 6, line 1 according to comments from reviewers 1 and 2.

4. Native check was again received.

We appreciate your consideration of the revised version.

Yours sincerely,

Kazumasa Shindo, M. D.
Department of Neurology
University of Yamanashi
1110 Shimokato, Chuo-city, Yamanashi 409-3898
Japan